Author's response to reviews

Title: An association between hOGG1Ser326Cys polymorphism and the risk of bladder cancer in non-smokers: a meta-analysis

Authors:

Changwei Ji (dr.jichangwei@gmail.com)
Zhao Liu (dr.zhliu@hotmail.com)
Huimei Chen (chenhuimei@nju.edu.cn)
Hongqian Guo (dr.guohongqian@gmail.com)
Changjian Liu (dr.cjliu@hotmail.com)

Version: 2 Date: 16 February 2012

Author's response to reviews: see over
Dear Editor,

Thank you very much for your mail about our manuscript entitled “An association between hOGG1Ser326Cys polymorphism and the risk of bladder cancer in non-smokers: a meta-analysis” (Manuscript No.: 1905621192676155). We thank the editors very much for the suggestion on the manuscript construction.

The followings are point by point response to the editor’s comments and suggestions.

1) Copy Editing - After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further.

   According to this comment, we polished the manuscript under the help of some native English speakers, especially in the abstract. We are very sorry about lots of errors in the previous vision, because of our carelessness. We hope this one will be qualified to be acceptable.

2) Please adhere to MOOSE Guidelines.

   We have downloaded the MOOSE Guidelines. After reading through the paper and checking our manuscript, we modified our manuscript in details. In our opinion, the new vision may adhere to the MOOSE Guidelines.

We are very grateful for your time and effort on the publications of our manuscript. If there is any other problem, please let me know. Thank you very much!

Sincerely yours,

Changjian Liu
Professor of Department of Vascular Surgery,
Medical School Affiliated Drum Tower Hospital, Nanjing University,
Nanjing 210008, China